Author's response to reviews

Title: Giardiasis in children living in post-earthquake camps from Armenia (Colombia)

Authors:

- Fabiana Lora-Suarez (Fabisuarez6@hotmail.com)
- Carolina Marin-Vasquez (Carola93@hotmail.com)
- Nelsy Loango (nelocha@hotmail.com)
- Martha Gallego (cibm@unicindio.edu.co)
- Elizabeth Torres (elito@hotmail.com)
- Maria M Gonzalez (mome90@hotmail.com)
- Jhon C Castano-Osorio (yiyma@yahoo.com)
- Jorge E Gomez-Marin (jegomezmarin@hotmail.com)

Version: 6 Date: 8 Mar 2002

Dear Editor BMC Public Health:

We have taken in account all recommendations and observations of Dr. Greig, their observations have contributed greatly to quality of paper.

Discretionary revisions:

All language suggestions were accepted and incorporated in the text

Compulsory revisions

To explain the differences in table 2 between first version and second and third revised version it would be taken in account that number of tables changed after first revision

The calculations of OR were remade using Statcalc in Epiinfo. The calculations were previously made by using Analysis of Epiinfo that is based on data sheet. In this new version all calculations were remade by using Statcalc and exact CI.

Table 3 was recalculated and OR for children not at school was revised. The paragraph in results and discussion sections were modified and Mantel Haenszel calculation included.

We hope these changes will make the paper now suitable for publication.

Yours sincerely,

Jorge E. Gomez-Marin MD, PhD